Author's response to reviews

Title: A young female with massive hemoptysis and deep venous thrombosis: HUGHES-STOVIN SYNDROME.

Authors:

Hamdan H AL-Jahdali (jahdali@yahoo.com)

Version: 2 Date: 4 March 2009

Author's response to reviews:

The requested revision:

Quarry: 1 Abstract should include introduction, case presentation, and conclusion as per journal style

Response: 1- Introduction rewritten to include introduction, case presentation, and conclusion

Quarry: 2 Include a competing interests section. If the authors have no competing interests, please state: "The authors declare that they have no competing interests."

Response: 2 DONE

Quarry: 3. Include a consent section at the end of the manuscript, before the reference list. This should contain a statement to confirm that informed written consent was received for publication of the manuscript and figures.

Response: 3 DONE

Quarry: 4 Reduce the reference list, as it is journal policy to only allow a maximum of 20 references.

Response: 4 Reduced to 20